Author's response to reviews

Title: Socio-demographic inequalities across a range of health status indicators and health behaviours among pregnant women in prenatal primary care: a cross-sectional study

Authors:

- Ruth Baron (r.baron@vumc.nl)
- Judith Manniën (judithmannien@gmail.com)
- Saskia te Velde (s.tevelde@vumc.nl)
- Trudy Klomp (Trudy.Klomp@Inholland.nl)
- Eileen K Hutton (huttone@mcmaster.ca)
- Johannes Brug (j.brug@vumc.nl)

Version: 3
Date: 16 July 2015

Author's response to reviews: see over
Dear Editor,

Thank you for reviewing and accepting our manuscript ‘Socio-demographic inequalities across a range of health status indicators and health behaviours among pregnant women in prenatal primary care: a cross-sectional study’ for publication. We have addressed the editorial points one by one below. We hope that all the editorial points and the journal criteria have been sufficiently addressed to enable production.

Kind regards,
Ruth Baron

Editorial points and authors’ responses:

- Please detail ethical approval within the methods section rather than at the end of the document.
  
  Authors’ response: We have added the line on ethical approval to the methods section. (lines 78 & 79)

- Please remove the section Funding and detail funding in the Acknowledgments section
  
  Authors’ response: We have removed the section on Funding. As there was no external funding for this study, this line has been omitted completely.

- Please rename the sections Disclosure of Interests and Contributions to Authorship as Competing Interests and Authors’ Contributions, respectively.
  
  Authors’ response: We have renamed the headings as Competing Interests and Authors’ Contributions respectively.

- Please provide the title and legend for figure 1 at the end of the document in the manuscript text and remove this information from the figure itself
  
  Authors’ response: We have removed the title and description from Figure 1 and added these to the end of the manuscript. (Page 19)